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Data Taking Status

● Last week: Monday March 29th  – Sunday April 4th

 Delivered 7742nb­1 Recorded 6320nb­1 Efficiency 81.6%

● This week: Monday April 5th  – Sunday April 11th

 Delivered 10109nb­1 Recorded 8861nb­1 Efficiency 87.7%           (4th best)

Day ∫L dt (del)   ∫L dt (rec) Efficiency
Mon Apr   5          1532nb­1             1252nb­1         81.7%
Tue Apr   6    1368nb­1             1224nb­1         89.4%
Wed Apr   7 1245nb­1    1075nb­1         86.4%
Thu Apr   8 2022nb­1    1848nb­1         91.4% 3rd best
Fri Apr   9        1283nb­1    1118nb­1         87.1%
Sat Apr 10        1500nb­1    1278nb­1         85.2%
Sun Apr 11        1159nb­1    1066nb­1         92.0%
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Store 3366 (Fri/Sat)

No significant interruptions of data taking
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Last Night's Store

p Halo

pbar Halo

Unable to take
physics data

Although they are
rarely exceeded

in “normal” stores,
we do have

operating limits

protons:
60kHz (warning)
80kHz (HV stdby)

antiprotons:
5kHz (warning)
7kHz (HV stdby)
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Beam Spot

(within specifications; yz­tilt close to limit)

300m
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Tracking / SMT

● Wednesday/Thursday: fewer SMT hits on tracks, broadened dca distribution etc.

NHits

dca

KS

● Resolution: new monitoring tools did 
not take solenoid polarity into 
account...
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Issues

Forward Muon MDT crate x30
About once a shift during physics data taking, timing errors from all front­ends reported, “crash” crate x30 – 

lost ~3 hours last week

Problem still being investigated, quick recovery procedure (<1min) in place

Level 1 Muon Power Supply Trips

~5 times during the week – inspected on Sunday

Fiber Tracker (CFT) LVPS

5.5V PS operated at 6.2V  change in wiring 
successfully reduced voltage drop

CFT front­end board

Failed Sunday morning during store – replaced 
between stores

L1CAL hot towers

Trigger Control Computer glitch
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Summary

Taking good data with 88% efficiency


